Author's response to reviews

Title: Trial of an educational intervention on patients' knowledge of atrial fibrillation and anticoagulant therapy, INR control, and outcome of Treatment with warfarin (TREAT)

Authors:

Danielle E Smith (danielle.smith@swbh.nhs.uk)
Christian Borg Xuereb (christian.borgxuereb@swbh.nhs.uk)
Helen M Pattison (h.m.pattison@aston.ac.uk)
Gregory YH Lip (g.y.h.lip@bham.ac.uk)
Deirdre A Lane (deirdre.lane@swbh.nhs.uk)

Version: 4 Date: 17 May 2010

Author's response to reviews:

17 May 2010

MS: 1868711027333689

Trial of an educational intervention on patients' knowledge of atrial fibrillation and anticoagulant therapy, INR control, and outcome of Treatment with warfarin (TREAT)

Danielle E Smith, Christian Borg Xuereb, Helen M Pattison, Gregory YH Lip and Deirdre A Lane

Dear BioMed Central

We would like to thank the Reviewers for their comments. We have revised the manuscript and made all of the formatting changes suggested. We hope that the paper is now ready for publication in BMC.

We look forward to hearing from you.

Kind regards

Danielle Smith